EO-1 Weekly Status Week of September 21 – September 27, 2007

Day of Year 264 - 270
Mission Day 2494 - 2500
Earth Observing-One (EO-1) General:

There were 111 Data Collection Events (DCEs) scheduled this week.  

An all instrument lunar calibration was performed on 09/27/2007 (07-270) 07:17z.

Instruments underwent decontamination cycling as follows:

The Hyperion hyperspectral imager (HSI) performed a deicing from 09/24/2007 at 00:24z (07-267) to 09/24/2007 at 15:10z (07-267).  

The Advanced Land Imager (ALI) performed an outgassing from 09/24/2007 at 00:32z (07-267) to 09/24/2007 at 15:24z (07-267).
The GSFC EO-1 team has been coordinating with NASA HQ to initiate a de-orbit waiver.  The plan is to stop lowering the orbit and then to continue to do inclination burns to maintain the Mean Local Time (MLT) at 10:00 am or later.  This will assure that the Hyperion image data will maintain scientific value.   

EO-1 Spacecraft Subsystems:

All spacecraft subsystems functioned nominally with the exception of the Command & Data Handling (C&DH) telemetry playback anomaly reported previously.  

Instruments

All instruments operated nominally this week.  

Technology Activities:
In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations

A teleconference was held between the Advanced Information System Technology (AIST) Sensor Web collaborators on Thursday, September 27.  The following items of interest were discussed:
· The need for the University of Alabama/Huntsville (UAH) to receive not only Level 1R Hyperion image data but also housekeeping metadata to permit a SensorML workflow to be constructed.  Also, they will be sent a Katrina flood Hyperion Level 1R data to be used to validate the generation of their flood classifier.  With this information, UAH has offered to implement a SensorML workflow in their client to generate a geolocated Hyperion product and then run a flood classifier on it. 
· The need to identify a server that will be used to routinely house new Level 1R data and housekeeping metadata for the SensorML workflow.    
· The need to re-host the SensorML client workflow and SensorML workflow execution engine as web service at GSFC.
The collaborators supported another UAS wildfire demo flight on Thursday, 27 September.

The GEOSS wildfire and volcano scenario videos have been produced as part of an overall initial operating capability (IOC) for GEOSS.  The IOC description is contained on an OGC password protected website for final review by the Pilot participants prior to being “mastered” into a kiosk demo package.  This package will be presented in an exhibit hall as part of the GEO summit III in November 2007 in South Africa.

Addition of KML File-Generation to EO-1 Operations:  No change.
Autonomous Sciencecraft Experiment (ASE)

ASE controlled EO-1 all week.

Ground System

Hardware Status

	Hardware
	Status
	Comments

	Primary R/T System
	
	Green
	

	Backup R/T System
	
	Green
	No change

	bMOC at DataLynx
	
	Green
	

	Playback System
	
	Green
	

	Mission Planning
	
	Yellow
	No change.

	Flight Dynamics
	
	Green
	FOT continued to perform re-entry analysis on the spacecraft.

	DTAS Server
	
	Yellow
	No change.

	ANS
	
	Green
	

	Flight Software Lab
	
	Green
	

	Admin System
	
	Green
	


Operations

New Lunar Calibration Sequence

FOT will be re-visiting the lunar calibration sequence.  A new sequence will be designed to include the fifth scan as a Hyperion centered scan.  This way, the first four scans will be centered with the ALI instrument’s four SEAs and the fifth scan will be centered with the Hyperion instrument.  Critical slew settling times will have to change once again to account for the added scan.  Affects of this change will be analyzed and tested thoroughly by the FOT before implementing the new sequence.

New Lamp Calibration Sequence

FOT will be testing a new flood lamp calibration sequence.  Flatsat will be utilized as much as possible during the testing of this new sequence.  Timing of the commands to perform the calibration will be studied.  In this new sequence FOT will try to enable the data gates prior to turning on the flood lamps.  An ATS load will be up-linked to Flatsat and FOT will monitor the execution of the commands.  Listed below is the command sequence that is executed each time /I_FL_CAL command is issued to the SC,

I_LAMPENABLE          enables lamp power
I_LAMPON oxb -1         turn on all lamps
Wait 10 seconds
I_SET_FPE_DG -1         turns on the FPE data gate
Wait two seconds
I_LAMPON 3 -1             turn off one lamp
Wait two seconds
I_LAMPON 1 -1             turn off one lamp
Wait two seconds
I_LAMPRESET              turns off all lamps
Wait two seconds
I_CLR_FPE_DG             turns off the FPE data gate

FOT will build a new command load to bring the /I_SET_FPE_DG -1 command to the beginning of the command sequence.  This way the data gates will be on before the lamps are powered on.

Re-Entry Analysis of EO-1

A change to the de-orbit profile has been requested by the EO-1 Mission Science Office.  The essence of the change is to allow the EO-1 to begin inclination adjustments rather than orbit lowering adjustments to maintain the MLT at later than 10:00 am and which will leave the altitude where it is now.  This orbit profile would delay EO-1’s reentry date.
New Antenna Additions to EO-1

EO-1 FOT has performed another test with the Norway station to certify the SG2 antenna.  This test was performed on September 25, 2007.  Due to some technical misunderstandings, EO-1 was unable to complete the connectivity test with SG2.  Another connectivity test will be scheduled in two weeks.

Use of EO-1 for Solar Dynamics Observatory (SDO) Antenna Tracking Certification
EO-1 will be providing two downlinks to the SDOGs antennas on Friday, September 28, 2007.  FOT will continue to support SDOs efforts to check-out their ground stations.
New Discrepancy Reports (DR)

There are no new requests this week.

New Enhancement Requests (ER)
There are no new requests this week  

New Operations Database Change Requests (DBCR)

There are no new requests this week.

Ground and Space Network

EOS Data and Operations System (EDOS), GSFC, MD, USA

There are no major problems to report.
McMurdo Ground Station (MGS), McMurdo, Antarctica

There are no major problems to report.

Wallops Ground Station (WGS), Wallops Island, VA, USA

There are no major problems to report.

DataLynx Ground Station (PF1), Poker Flat, AK, USA

There are no major problems to report.

Alaska Ground Station (AGS), Poker Flat, AK, USA 

There are no major problems to report.
Svalbard Ground Station (SGS), Longyearbyen, Norway

There are no major problems to report.

Tracking and Data Relay Satellite System (TDRSS), White Sands, NM, USA

There are no major problems to report.
USGS Multi-Satellite Ground Station (LGS), Sioux Falls, SD, USA
LGS is now being scheduled only for contingencies.

Tasmanian Earth Resources Satellite Station, a.k.a. Hobart (HGS), Hobart, Australia

There are no major problems to report.
Cordoba Ground Station (CGS), Cordoba, Argentina

No passes were scheduled this week. 

Upcoming Events:

No events are scheduled for the next two weeks.
Imagery Status:

Scenes and Engineering Cals planned for week of Sept 21 – Sept 27, 2007                          111
Total scenes and engineering calibrations planned for entire mission – approximately 36,298
Total Scenes:  ALI scenes in the Level 0 archive            32,833 (as of Sept 27, 2007)

                        Hyperion scenes in the Level 0 archive    32,583
Publications and Presentations Status (as of 07-05-07)
404 publications 

284 external presentations

51 articles and press releases
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